

Effective Disclosures in Financial Decisionmaking

Angela A. Hung, Min Gong, Jeremy Burke

For more information on this publication, visit www.rand.org/t/RR1270

Published by the RAND Corporation, Santa Monica, Calif.

© Copyright 2015 RAND Corporation

RAND® is a registered trademark.

Limited Print and Electronic Distribution Rights

This document and trademark(s) contained herein are protected by law. This representation of RAND intellectual property is provided for noncommercial use only. Unauthorized posting of this publication online is prohibited. Permission is given to duplicate this document for personal use only, as long as it is unaltered and complete. Permission is required from RAND to reproduce, or reuse in another form, any of its research documents for commercial use. For information on reprint and linking permissions, please visit www.rand.org/pubs/permissions.html.

The RAND Corporation is a research organization that develops solutions to public policy challenges to help make communities throughout the world safer and more secure, healthier and more prosperous. RAND is nonprofit, nonpartisan, and committed to the public interest.

RAND's publications do not necessarily reflect the opinions of its research clients and sponsors.

Support RAND

Make a tax-deductible charitable contribution at www.rand.org/giving/contribute

www.rand.org

Preface

The work reported here was undertaken in 2015 and was sponsored by the Department of Labor (DoL). The report should be of interest to DoL staff; staff of other federal agencies that have regulatory responsibilities related to financial markets, such as the U.S. Securities and Exchange Commission; broker-dealers who provide advisory services related to individual retirement accounts; and economists and policy analysts with interests related to potential effects of conflicts of interest for financial advisors and potential policy responses.

This research was undertaken within the Center for Financial and Economic Decision Making (CFED). The mission of CFED, a part of RAND's Labor and Population research division, is to understand how people in the United States and around the world collect and think about financial information and how successfully they match their financial decisions to their interests and goals. CFED's researchers are dedicated to finding solutions that can improve the decisionmaking that affects the financial well-being of individuals, families, and nations. RAND Labor and Population has built an international reputation for conducting objective, high-quality, empirical research to support and improve policies and organizations around the world.

For more information on RAND Labor and Population, contact Unit Director, RAND Labor and Population, 1776 Main Street, P.O. Box 2138, Santa Monica, CA, 90407–2138; call (310) 393-0411; or visit the RAND Labor and Population homepage at http://www.rand.org/labor.

Contents

PrefacePreface	iii
1. Introduction	1
2. Use of Disclosures of Conflicts of Interest in the Financial Industry	2
Disclosure in International Financial Advice Markets	
3. Effectiveness of Disclosures of Conflicts of Interest	5
Theoretical Models on the Impact of Disclosure of Conflicts of Interest	5
Impact of Disclosure of Conflicts of Interest on Consumers	6
Impact of Disclosure of Conflicts of Interests on Advisers	8
Discussion	9
4. Effectiveness of Other Disclosures in the Financial Services Industry	11
Consumer Mortgage Disclosures	11
Credit Card Disclosures	13
Mutual Fund Disclosures	15
Discussion	18
5. Best Practices in Designing Effective Disclosures	19
Guidelines from Regulations	19
The Clear and Conspicuous Standards of the Federal Trade Commission	19
SEC's Plain English Initiative	19
FTC's .Com Disclosure Guidelines	20
Regulatory Guideline on the Timing of Disclosure	20
Lessons Learned from Effective and Ineffective Disclosures	20
Discussion	22
6. Summary	24
References	25

1. Introduction

In the financial services market, financial service providers often have better information about the quality, features, fees, risks, and benefits of their products or services than consumers. In a market with this type of asymmetric information, disclosure is an often-used policy tool designed to increase transparency and provide consumers with valuable information to make informed decisions. In this report, we define disclosure as a statement that provides relevant information to consumers for informed decisionmaking. A disclosure often reveals estimated costs and impacts to consumers, commitments of the relevant parties, existence of any conflicts of interest, and the nature of the relationship between parties.

Disclosures are ubiquitous in the financial industry. For example, the Investment Advisers Act of 1940 requires investment advisers to disclose potential conflicts of interest to clients. The Truth in Lending Act (TILA) of 1968 requires disclosure of terms and costs of consumer credit to potential borrowers. Even though disclosure is a commonly used regulatory tool, it is an open question whether disclosures are effective in reducing information asymmetry and improving consumer decisions. In theory, once relevant information is disclosed to consumers, they can then make informed decisions. However, in practice, consumers may have limited attention or limited understanding of the disclosure. Furthermore, they may not have the ability to appropriately take the disclosed information into account as they make financial decisions.

In this report, we review the literature on consumer disclosures in the financial industry. The primary focus of the review is on disclosures of conflicts of interest, particularly with regard to financial advice, but we also examine use of disclosure associated with other common financial products or services, including credit cards, mortgages, and mutual funds. In particular, we are interested in the following questions:

- 1. Can disclosure be effective in
 - a. conveying desired information in a way that the recipient can understand it,
 - b. providing decision support, and
 - c. helping consumers make decisions aligned with their own interests?
- 2. In what instances is disclosure most effective?

The next chapter discusses use of disclosure on conflicts of interest in the U.S. financial advice market, as well as financial advice markets in other countries. Chapter 3 reviews the research literature on effectiveness of disclosure on conflicts of interest. Chapter 4 reviews the literature on other disclosures in the U.S. financial industry, such as mortgage, credit card, mutual fund, and payday loan disclosures. Chapter 5 reviews the literature on best practices for effective disclosures, and Chapter 6 concludes.

2. Use of Disclosures of Conflicts of Interest in the Financial Industry

Conflicts of interest in the financial industry have been a key policy issue, as evidenced by President Obama's February 2015 call for new regulatory action on the issue of conflicts of interest in retirement advice. Conflicts of interest in financial advising can arise when the adviser has incentives that are not necessarily aligned with the client's best interests. Common conflicts of interest include "dealing with affiliates, the receipt of compensation or other benefits from third parties that may affect the independence of the advice provided, an adviser's financial interest in a transaction (e.g., acting as principal), client referral arrangements and personal and proprietary trading by the investment adviser and its employees" (Klass, 2008).

Conflicts of interest can result in biased advice. For example, a conflicted adviser might be more likely to recommend certain financial products if she receives commissions or fees from selling them, or to encourage active trading if her compensation is positively related to the number of transactions, even if such behavior is not in the best interest of her client. Biased advice can have significant negative impacts on investors. The President's Council of Economic Advisers recently estimated that conflicted advice reduces returns to retirement savings by 1 percentage point, costing Americans an estimated \$17 billion per year based on an estimated \$1.7 trillion of IRA assets (Council of Economic Advisers, 2015).

In the United States, investment advisers and broker-dealers provide financial advice to retail investors. Despite the overlap in services, brokers and investment advisers are subject to different federal regulations. Because of the different regulatory regimes for investment advisers and broker-dealers, they are subject to different standards when providing investment advice. Broker-dealers are obligated to make suitable recommendations. That is, a broker-dealer making a recommendation to a retail customer must have grounds for believing that the recommendation is suitable for that customer with respect to his or her portfolio, financial situation, and needs. Broker-dealers may also have additional suitability requirements, depending on the products that they offer. Unlike broker-dealers, federally registered investment advisers owe a fiduciary obligation to their clients. These obligations require the adviser to act solely with the client's investment goals and interests in mind, free from any direct or indirect conflicts of interest that would tempt the adviser to make recommendations that would also benefit him or her. Furthermore, investment advisers are required to disclose any potential conflicts of interest.

¹ The Securities Exchange Act of 1934 (48 Stat. 881) regulates brokers and dealers, and they are also subject to oversight from FINRA, an independent self-regulatory organization. The Securities and Exchange Commission (SEC) is the regulatory body for investment advisers, under the Investment Advisers Act of 1940 (54 Stat. 847).

Disclosure in International Financial Advice Markets

As in the United States, disclosure is a key feature of regulatory regimes in financial advice markets around the globe. As part of its Retail Distribution Review (RDR), which banned financial advisers from receiving commission, the United Kingdom (UK) required "restricted" advisers (those who can recommend only certain products, product providers, or both) to disclose the nature of their restriction to clients (Burke and Hung, 2015). Financial advisers are also required to disclose to prospective clients up front how much the advice will cost and the form in which it will be paid.²

Similar to the RDR in the UK, Australia recently implemented legislation to harmonize adviser and advisee incentives by banning commission payments through its Future of Financial Advice (FoFA) reforms. A key provision of FoFA is the requirement that advisers provide clients who have entered into an ongoing fee arrangement with an annual fee disclosure statement which describes (1) "the amount of each ongoing fee paid by the client in the previous year," (2) "information about the services that the client was entitled to receive in the previous year," and (3) "information about the services that the client received in the previous year" (Australian Securities and Investments Commission, 2013).

Disclosure often takes on an even more central role in regulatory regimes where commission is still permitted. In Germany, for example, advisers are legally required to disclose to clients any inducements received.³ Advisers are also required to provide clients with a product information sheet for each investment the client is advised to purchase. The product information sheet should contain all the information required for an investor to make an informed comparison across financial instruments, including the nature of the recommended financial instrument, how it works, and its associated costs and risks. To improve readability, the information sheet must be no longer than two or three pages and written in a clear way. Importantly, the information sheet must be provided in a "timely manner" before a contract on a transaction is executed (Federal Financial Supervisory Authority, 2010).

Financial advisers in Singapore have similar disclosure requirements to those in Germany. In particular, financial advisers in Singapore must disclose to a client all compensation that the adviser will receive that is directly related to any product recommendations (or execution of recommended transactions) including any commission, fee, or other benefits. Advisers must also describe key features of any product that is recommended, including the nature of the product, the client segment for whom the product is intended, details on the product provider (including any relationship between the adviser and the provider), and the associated benefits, risk, and costs (Monetary Authority of Singapore, 2002).

3

² Financial Conduct Authority (FCA), 2015.

³ BaFin, undated.

While disclosure is an important component of many regulators' strategy to improve transparency in financial advice markets, advisers may not always follow those requirements. Following the introduction of the RDR, the UK's FCA conducted a series of thematic reviews to investigate how firms were adapting to the new regulatory environment. The second of the three planned reviews focused on (1) whether firms describing themselves as independent were operating independently in practice, and (2) how firms were disclosing their service proposition and the associated charges to clients. Of the 88 firms reviewed that claimed to offer independent advice, the FCA identified two not acting independently in practice and had doubts about an additional 28 firms. FCA then reviewed 17 of the 28 firms in more depth and concluded that six were not operating independently; there were concerns that another four firms might not be acting independently, but FCA did not have sufficient evidence to determine this with certainty (FCA, 2014a). In regard to cost disclosure, the FCA found that a high proportion of firms were failing to correctly disclose to clients the cost of advice, the type of service offered, and the nature of any ongoing service. Of the firms surveyed, 73 percent failed to provide the required generic information regarding how they charge for advice and/or did not clearly convey the cost of advice to clients in a timely manner (FCA, 2014b). However, these inadequacies may have been largely due to uncertainty surrounding the recently imposed requirements. The FCA's third thematic review found that firms had made significant improvements in how costs and the scope and nature of services were disclosed, though concerns remained that some firms were failing to clearly disclose, in cash (pounds) terms, the ongoing fees clients would be charged for ongoing service (FCA, 2014c).

Relatedly, the Monetary Authority of Singapore (MAS) conducted an audit study in 2011 to examine the quality of financial advice provided by advisers and to investigate whether they were complying with regulation. MAS enlisted 126 mystery shoppers, who made 500 visits to 11 licensed banks and four registered insurance companies to receive financial advice. While advisers typically provided information on the nature and objective of the recommended product (93 percent of visits), the amount of fees and charges was not disclosed in 48 percent of visits, and the frequency of fees and charges was not disclosed in 58 percent of visits (MAS, 2012).

3. Effectiveness of Disclosures of Conflicts of Interest

This chapter reviews the literature on disclosure as a regulatory tool to increase transparency to help investors make informed decisions when facing advisers with conflicts of interest. We first review theoretical models on the impact of disclosure of conflicts of interest. We then review empirical studies examining the impact of disclosure of conflicts of interest on both consumers and advisers.

Theoretical Models on the Impact of Disclosure of Conflicts of Interest

Inderst and Ottaviani have developed a body of theoretical work on conflicts of interest in advising (Inderst and Ottaviani, 2012a, 2012b, 2012c).

They developed a model in which consumers receive advice about product suitability from advisers. A product is suitable for a consumer if it is the optimal choice to satisfy the consumer's specific needs. Advisers want to pair consumers with the appropriate product, either due to professional concerns, reputational cost, and/or fear of prosecution. However, in the model, advisers receive a commission⁴ if consumers purchase a particular product, and this product is not necessarily suitable for all consumers. Quality of the advice to consumers depends on both the potential bias (resulting from commission payment) and the effort level that the adviser expends to acquire information on the customer's specific circumstances and needs to help determine the suitability of the product.

In their basic model, there are two types of consumers, naïve and wary. Naïve consumers are not aware of commissions, or at least not aware of the potential impact of commissions on the quality of advice, while wary consumers are aware of commissions and take the adviser's incentives into account when evaluating advice. The model predicts that if there are only naïve consumers in the market, then consumers will not pay for advice directly but instead will pay for advice indirectly through higher product prices and receive biased advice. In this case, disclosure can turn naïve consumers into wary ones. The authors also compare mandatory disclosure with prohibiting or capping commissions and conclude that prohibiting commissions may have the unintended negative side effect of reducing the adviser's incentive to acquire information, while disclosure can protect naïve consumers and improve their welfare. Intuitively, when an adviser faces complex and specialized products, and effort is required to acquire information on which products suit the consumer's needs best, commission is an incentive to encourage the adviser to expend more effort to identify suitable products for the consumer.

5

⁴ In the model, advisers also receive a flat fee paid by the consumer as well as a fixed payment from the product provider.

Impact of Disclosure of Conflicts of Interest on Consumers

As described in the previous section, disclosure of conflicts of interest, in theory, can help investors make more informed judgments and decisions. However, there is concern from policymakers about how consumers react to disclosures of conflicts of interest. As articulated in the U.S. Department of Treasury's 2009 call for financial regulatory reform, "Consumers, however, may retain faith that the intermediary is working for them and placing their interests above his or her own, even if the conflict of interest is disclosed. Accordingly, in some cases consumers may reasonably but mistakenly rely on advice from conflicted intermediaries" (U.S. Department of the Treasury, 2009, p. 68).

The concern raised by the Treasury Department has been supported by empirical evidence. For example, Chater, Huck, and Inderst (2010) find, in an online experiment with 6,000 recent purchasers of retail financial services, that simply disclosing advisers' compensation schemes does not affect consumers' willingness to pay for or follow advice from advisers with a potential conflict of interest. Including a more strongly worded statement that the adviser's incentives might not be aligned with those of the consumer decreased willingness to pay for advice, but the vast majority of consumers still opted to pay for the potentially conflicted advice.

Cain, Loewenstein, and Moore (2005) argue that there are several reasons disclosure of conflicts of interest may not be an effective policy tool. Firstly, the behavioral impacts of conflicts of interest on advice are a complex problem that most consumers do not understand, and therefore, it is easy for consumers to underestimate how advice will be affected by conflicts of interest. Secondly, they argue that people have a tendency to "be naturally trusting and credulous toward their own advisers" (p. 5). As an example, they cite research from Gibbons et al. (1998), in which approximately 200 patients were surveyed, and 54 percent of respondents were aware of the existence of pharmaceutical gifts to doctors, but only 24 percent thought their own doctors accepted such gifts. Lastly, Cain, Loewenstein, and Moore (2005) argue that even if consumers understand that conflicts of interest may affect the advice that they receive, they typically do not make sufficient adjustments to their own behavior to offset the biased advice that they receive.

In experimental settings, Cain, Loewenstein, and Moore (2005 and 2011) found evidence that even though advisees might discount advice that they know comes from a conflicted source, advisees do not sufficiently discount the biased advice. In two different experiments, an adviser with private information provided advice to an advisee regarding the value of an object (a jar of coins in Cain, Loewenstein, and Moore, 2005, and sale prices of houses in Cain, Loewenstein, and Moore, 2011). The advisee's payoff depended on how accurately she estimated the value, but the adviser's payoff depended on the experimental treatment. In the no-conflict condition, the adviser's incentive was based on the advisee's accuracy. In two conflicted conditions, one in which the conflict was disclosed and one in which it was not, the adviser's incentive was based on the extent to which the advisee overestimated the value of the object (either the jar of money

or the home sale price). Both studies found that the difference between the advice given and the estimates made by the advisee was significantly greater when there was a disclosed conflict of interest than when there was an undisclosed conflict of interest or when interests were aligned, which suggests advisees do discount conflicted advice when the conflict is disclosed. In Cain, Loewenstein, and Moore (2011), for example, when the conflict of interest was not disclosed, advisees discounted the advice on the property value by \$11,000, or less than 5 percent of the adviser's recommendation, possibly because the advisees suspected a conflict of interest after receiving feedback on the actual values in the previous round. When the conflict of interest was disclosed, advisees discounted their estimates, on average, by about \$25,000, or about 10 percent of the adviser's recommendation. However, both papers also indicate that advisees do not sufficiently discount the conflicted advice. For example, in Cain, Loewenstein, and Moore (2011), advisers in the disclosed conflict condition inflated their advice by more than \$51,000, on average. Cain, Loewenstein, and Moore (2011) suggest two major reasons for the insufficient discounting. First, the conflicted advice can serve as an anchor that biases the decision toward the recommendation. Second, people are unlikely to know how to incorporate the disclosed information about the conflict of interest in their decision process and adjust their decisions appropriately. To do so, people need to possess "a wide range of subsidiary judgments, including the ethicality of the adviser, whether the adviser is a 'restrainer' or an 'exaggerator,' the cost of getting a second opinion, one's knowledge of the subject, and one's relationship with the adviser" (Cain, Loewenstein, and Moore, 2011). The existence of such a comprehensive information asymmetry between the adviser and advisee is unlikely to be sufficiently addressed by the disclosure alone.

Other experimental evidence suggests that disclosure of conflicts of interest may cause the unintended consequence of adherence to biased advice. For example, disclosure can lead to an increase in trust when interpreted as a sign of honesty (Pearson et al., 2006) or to a decrease in perceived risk when regarded as an indication of governmental protection that "they are being looked after, causing them to become less vigilant" (Green and Armstrong, 2012).

Sah, Loewenstein, and Cain (2013) describe a phenomenon that they refer to as "burden of disclosure," in which advisees feel increased pressure to adhere to advice once a conflict of interest is disclosed, even though they may distrust the advice. In an experimental setting, they found that 53 percent of advisees who received bad advice from an adviser with an undisclosed conflict of interest followed the advice that they were given, but 81 percent followed the bad recommendation when the conflict of interest was disclosed. When advisers gave bad advice, advisees who received disclosure were significantly less happy with their choice and were more likely to report that the adviser put his own interests first. At the same time, advisees who received disclosure were significantly more likely to report that they felt pressure to help their adviser and that they were uncomfortable rejecting the advice. Lowenstein, Sah, and Cain (2011) argue that the burden of disclosure is driven by two mechanisms: insinuation anxiety, in which advisees do not want to reject advice for fear that the adviser will think that the advisee finds her

to be corrupt; and the panhandler effect, in which advisees feel pressure to make a decision that financially benefits the adviser once the adviser's compensation structure becomes common knowledge.

Unlike the studies above, one study found that disclosing advisers' incentives caused investors to be less likely to invest when they knew it would result in higher commissions for advisers. Chater, Huck, and Inderst (2010) conducted a laboratory experiment in which 484 participants from three European Union countries (Czech Republic, Germany, UK) play an investment game with advice either from a conflicted adviser or from a nonconflicted adviser. They find a significant impact of disclosing adviser incentives. In one task, advisees make a decision on how much they should invest in a risky investment. Advisees invest about 900 euros less when they have a conflicted adviser, paid based on their investment amount, than when their adviser is paid a flat fee. In another task, advisees need to allocate their funds to two investments, one superior to the other. When it is disclosed that their advisers have a conflict of interest, advisees trust the recommendations less, even when the recommendation is the optimal investment. Advisees invest almost 1,600 euros less in the optimal investment when a conflicted adviser recommends it than when a nonconflicted adviser recommends it.

Impact of Disclosure of Conflicts of Interests on Advisers

Up until now, we have been reviewing evidence on the impact of disclosure of conflicts of interest on consumer behavior. However, even if consumer behavior is unaffected by disclosure, consumer outcomes may be influenced if disclosure affects adviser behavior. Cain, Loewenstein, and Moore (2011) propose three different behavioral responses by advisers to disclosure. An adviser who expects that his advice might be discounted may inflate or exaggerate his advice, a behavior that Cain, Loewenstein, and Moore refer to as *strategic exaggeration*. Alternatively, an adviser may engage in *strategic restraint*: If he anticipates that his biased advice will be rejected once his conflicts of interest are disclosed, he may rein in or steer away from biased advice. Lastly, another possible behavioral response is *moral licensing*, in which disclosure allows an adviser to adjust his moral or professional standards. Moral licensing might occur, for example, if an adviser feels that he can rationalize providing biased advice because his conflicts of interest have been fully disclosed.

Both strategic exaggeration and moral licensing are supported by evidence from experimental studies. In the Cain, Loewenstein, and Moore (2005 and 2011) experiments described above, the authors found evidence that in the presence of disclosure, advisers with conflicted incentives provide more biased advice than in the absence of disclosure. Note that this inflated advice, together with insufficient discounting by advisees as described in the previous section, results in advisees being worse off (earning lower payoffs) when conflicts of interest are disclosed than when they are not disclosed. In another experiment in Cain, Loewenstein, and Moore (2011), subjects rated the ethicality of offering biased advice as "somewhat unethical"

when the advisee is unaware of the conflict of interest, but rated biased advice as "somewhat ethical" when the conflict is disclosed.

While the discussion above generally indicates that disclosure increases, rather than reduces, bias in advice with unavoidable conflicts of interest (such as in the case of financial advisers, since the compensation structure cannot be easily altered by the adviser to avoid conflicts), research has shown that disclosure can be beneficial when the conflict is avoidable. In a laboratory experiment, Sah and Loewenstein (2014) found that because people are averse to being viewed as biased, disclosure (either mandatory or voluntary) can discourage advisers from accepting conflicts of interest when avoidable, thereby improving the quality of provided advice. An example of avoidable conflict of interest is doctors who can decide whether to accept gifts from pharmaceutical companies. A report by the Department of Health of the District of Columbia suggests that disclosing the speaking fees received by physicians may discourage the accumulation of those fees in the future. It is possible that "physicians voluntarily decrease financial relationships with companies after public disclosure" or "companies decrease financial relationships with doctors after public disclosure" (George Washington University, 2012). The effectiveness of disclosure can also be improved if combined with other interventions. For example, Church and Kuang (2009) found that in an experimental setting where the investor had the ability (at a cost) to penalize bad advice in conjunction with disclosure, advisers offered less biased advice, and investor payoffs increased.

Discussion

In principle, disclosure can increase investor awareness of conflicts of interest, potentially mitigating their impacts. However, our review of existing studies indicates that disclosure of conflicts of interest may not improve outcomes for all consumers. When conflicts of interest are disclosed, many consumers do not know how to respond appropriately due to various factors, such as lack of a way to accurately estimate the adviser's bias in a recommendation, or the cost of searching for a second opinion. Many consumers fail to adjust their behavior sufficiently, if at all, when conflicts are disclosed. Disclosure can also cause unintended consequences:

Consumers may feel a "burden of disclosure" to follow the advice, and advisers may respond to the disclosure by providing even more biased advice, resulting in decreased welfare for consumers.

We would like to include a word of caution on the results reported here. The majority of existing research reviewed is based on either theoretical modeling or controlled experiment results. The external validity of the conclusions drawn from this research is an open question. For example, the monetary incentive, the cognitive load, and the decision environment are very different when people are estimating the value of a jar of coins as opposed to when they are making a financial decision regarding retirement.

In addition, many important questions remain unanswered. What percentage of investors actually read disclosures of conflicts of interest? To what degree do investors understand disclosures and the impact of conflicts of interest on adviser behavior? Are there systematic individual differences? In other words, can we characterize consumers least likely to understand disclosures of conflicts of interest? Additional research, with more diversified data sources and methods (such as in-depth interviews, focus groups, case studies, and field experiments), is needed to more fully evaluate the effectiveness of disclosure as a policy intervention to protect customers. However, there is considerable evidence suggesting that disclosure of conflicts of interest in isolation may not improve welfare for all consumers.

4. Effectiveness of Other Disclosures in the Financial Services Industry

In this section, we review literature on other disclosures used in the financial services industry, especially disclosures on fees, and how those disclosures impact consumer behavior and outcomes. In particular, we focus on disclosures associated with mortgages, credit cards, and mutual funds

Consumer Mortgage Disclosures

Consumers who are shopping for mortgages are faced with a complex decision problem that many of them, particularly first-time buyers, are unfamiliar with. To help individuals make better choices, terms and costs of credit must be disclosed to consumers to provide them with timely information during the loan application process.⁵

There are three primary disclosure documents for mortgage loans in the United States. The Truth in Lending Act (TILA) of 1968 requires that consumers be provided with a Truth-in-Lending disclosure. An initial TILA disclosure is required within three business days of loan application, and a final TILA disclosure is required at loan closing. The TILA statement discloses key terms and costs of the loan, including annual percentage rate (APR), finance charge, the total amount financed, and the total of payments. It also includes information on payment schedule, late payment fees, and prepayment penalties.

The second disclosure document is the Good Faith Estimate (GFE) required under the Real Estate Settlement Procedures Act of 1974 (RESPA). The GFE is "the lender's best estimate in 'good faith' of closing costs—that is, the one-time fees for services related to the completion of mortgage transaction" (Collins, 2011). The estimate must include an itemized list of fees and costs such as legal fees, title search fees, and document fees, and must be provided within three business days of the borrower's application.

The third disclosure document is the HUD-1 (or HUD-1a in refinancing) Settlement Statement, which is a companion to the GFE: GFE provides the estimated closing costs, and HUD-1 Statement lists the actual final closing costs. A preliminary copy of a HUD-1 Settlement Statement is required if the borrower requests it 24 hours before closing.

Lacko and Pappalardo (2007, 2010) find that mortgage borrowers display a lack of understanding of key terms in the TILA statement and GFE form, in part because of the design

_

⁵ While conflicts of interest may arise in the mortgage industry, we discuss consumer mortgages in this section rather than the previous sections because the research literature on disclosure in the mortgage industry focuses on disclosure of fees and costs rather than disclosure of conflicts of interest.

of those disclosures. The authors conduct in-person interviews with 36 mortgage borrowers who obtained a loan in the past four months. They also conduct an experiment with over 800 mortgage borrowers who obtained a loan in the past two and a half years. The experiment was designed to compare consumer understanding of the current TILA and GFE disclosures with prototype disclosures of the authors' design. They find that the TILA statement and GFE do not properly convey important information about loan terms and costs to consumers. From the interviews, they find that many respondents are confused by the disclosures and do not understand key terms such as the APR, amount financed, and discount fees. In fact, many respondents also do not understand the costs and terms of their own recently obtained mortgages. Experiment results confirmed the interview findings. For example, about a fifth of experiment subjects cannot identify the loan APR, the amount of cash due at closing, or the monthly payment. About one-third cannot identify the interest rate or which of two loans is less expensive. Two-thirds are unaware of the prepayment penalty. Almost nine-tenths cannot identify the total amount of up-front charges. The authors argue that consumers do have the capability to understand the loan terms and costs, but the current forms are poorly designed. They find that consumers who received the prototype disclosures display significantly greater understanding of the key terms and costs than those who received the current disclosures.

In a survey of 102 mortgage brokers, Sovern (2010, 2014) finds that the lack of impact of disclosures on borrower behavior might be due to limited attention to the disclosures. Two-thirds of the brokers report that less than 30 percent of their borrowers spend more than a minute reviewing disclosures. The brokers unanimously report that borrowers never withdraw from a loan after reading the final disclosures at the closing, and never use the disclosures for comparison shopping.

In July 2010, the Dodd-Frank Act was enacted and transferred rulemaking authority under both TILA and RESPA to the Consumer Financial Protection Bureau (CFPB). Under the mandate from the Dodd-Frank Act, CFPB issued proposed integrated disclosure forms and rules for public comment in July of 2012. The new disclosure forms were shown to perform significantly better than the current disclosure forms in terms of improving consumer understanding, allowing better comparison shopping, and avoiding costly surprises at the closing table, according to CFPB studies conducted with 858 consumers in 20 locations ("Integrated Mortgage Disclosures Under the Real Estate Settlement Procedures Act (Regulation X) and the Truth in Lending Act (Regulation Z)," 2013; CFPB, 2013a). On December 31, 2013, the 2013 TILA-RESPA Final Rule was published and is expected to take effect on October 3, 2015. The new Loan Estimate form combines the current early TILA statement and GFE and is required to be provided within three days of application, and the new Closing Disclosure form combines the current HUD-1 and final TILA statement and is required to be provided three days before closing.

Credit Card Disclosures

Credit card usage is ubiquitous in the United States. In 2009, there were 156 million credit cardholders with a total purchase volume of \$1.94 trillion (U.S. Census Bureau, 2012). The Federal Reserve's Survey of Consumer Payment Choice (Foster et al., 2011) reports that 72 percent of U.S. consumers are credit cardholders.

Credit card disclosure is regulated under the Truth in Lending Act of 1968 and the Fair and Accurate Credit Transaction Act of 2003 (FACT), among others. The laws regulate the contents, format, and timing of the credit card disclosures to ensure that consumers can understand the credit terms and shop for credit more readily and knowledgeably (CFPB, 2013b).

Durkin (2006) conducted surveys to explore consumers' knowledge about their credit card accounts, the perception and usage of credit card disclosures, and consumers' preferences regarding information to be conveyed by credit card disclosure. Almost 500 respondents were asked about disclosures included in credit card periodic statements. The majority of the consumers (62 percent) report that they check the APRs regularly (at least four times a year), but 66 percent of consumers read the disclosures on the back of the statements fewer than four or five times a year. Consumers who report a credit card balance of more than \$4,500 are more likely to check both the APRs and the back material frequently than those who carry a smaller balance or no balance at all. Approximately 40 percent of the respondents report that either the APR or the finance change has affected their card use decisions. The survey also finds that for those who have opened a new account in the past year, about three-quarters think that the information provided in the account-opening disclosures is useful. Among those who do not think the disclosures useful, most report that there is too much information or it is too confusing or legalistic.

In 2009, the Credit Card Accountability Responsibility and Disclosure (CARD) Act was launched to bring tighter regulation to all consumer credit cards with the purpose of increasing transparency and fairness in credit card contracts. One of the requirements of the CARD Act is that credit card statements need to show how long it will take cardholders to pay off their balances and how much consumers will pay in total if they pay the minimum payment only, in comparison to the total cost if they pay off the entire balance in 36 months. In its CARD Act Review report (CFPB, 2013c), CFPB states that a small minority of card issuers reports a reduction in the percentage of customers making only the minimum payment. This reported change is confirmed by Agarwal et al. (2014), in which the authors use a difference-in-difference approach to estimate changes in repayment behavior between January 2008 and December 2012 for consumer credit cards, which are subject to the CARD Act, and small business credit cards, which are not subject to the regulations. The results from the panel data of 160 million credit card accounts show that the new disclosure requirements have had a small but significant effect on borrowers' repayment behavior. The number of customers who pay at a rate to repay the balance within 36 months increased from 5.3 percent to 5.7 percent. The authors estimate that the

upper bound impact from the new disclosure requirements is \$57 million in annualized interest saving.

Jones, Loibl, and Tennyson (2012) estimate the impact of disclosure changes mandated by the CARD Act on consumers' repayment decisions based on monthly survey data from 300–500 households per month between June 2006 and December 2011. The authors find that the likelihood of a household paying off its most recent credit card bill in full significantly increased (with a magnitude between 3.8 percent and 4.8 percent) after the new disclosure. The probability of skipping a payment is estimated to be about 1 percent to 1.5 percent smaller after the CARD disclosure. However, they find minimal evidence on debt repayment behavior among those who continue to carry debts.

In another study, Salisbury (2014) examines the impact of minimum payment warnings on repayment behavior. She finds that disclosing information on the costs of paying only the minimum payment (both interest costs and long payoff duration) has no impact on repayment behavior for most respondents. However, when minimum payment information is accompanied by information on an alternative course of repayment that would pay off the balance in three years, consumers are significantly more likely to repay an amount close to the three-year repayment amount. However, she also identifies a possible "backfire effect" where people who would have paid more than the three-year amount are less likely to do so when the three-year amount is present. Both Navarro-Martinez et al. (2011) and Stewart (2009) find supporting evidence of this backfire effect. They find that people make larger credit card payments in the absence of minimum payment information on credit card statements.

There has been some suggestive evidence that the new statement requirements may have helped consumers better understand credit card terms. For example, one study from Soll, Keeney, and Larrick (2013) finds that in an online survey with 543 participants, without the minimum payment warning, less numerate people tend to underestimate the monthly payment required to pay off a debt in 36 months, whereas more numerate people tend to overestimate the monthly payment. When subjects were provided with the minimum payment warning, both overand underestimation were substantially reduced. The authors do point out, however, that even though the minimum payment warning statement is clear that the amounts given assume no further activity, subjects, particularly those with lower numeracy, still underestimate the size of required payment to pay off the loan in three years in the event that there is further activity. Another study by the authors also identifies additional confusion from the new statements. When asked how long it takes to pay off a balance if the consumer always pays only the current minimum payment in the new statement, only 7 percent of the respondents give the correct answer, while 49 percent of them answer the same question correctly with the old statement.

One reason for the relatively small effect of the minimum payment warning may be limited comprehension. J. D. Power's 2013 Credit Card Satisfaction Survey reports that only 47 percent of cardholders say they "completely" understand their credit card terms. Among those who do not fully understand the terms, 73 percent of them indicate confusion over interest rates, and 31

percent of them do not understand the late fee. Another reason may be that required disclosures are included with paper credit card statements, but more and more credit cardholders pay their bills online and may not be exposed to the disclosures. The CFPB (2013b) finds that no issuers provide the minimum payment warning on the payment screen in their online account websites, but issuers who track customer online payment behavior report that 38 percent of customers made at least one payment online, and only 26 percent of consumers opened their statements before paying online.

Lastly, even though payday⁶ loans are a different type of short-term loan from credit cards, research on payday loan disclosures provides some insight. Bertrand and Morse (2011) ran a randomized field experiment to test how payday loan borrowers reacted to disclosures. Customers who entered 77 stores in 11 states over a period of two weeks were given one of three carefully designed disclosures. The disclosure that compared the dollar cost of repeated borrowings via payday loans versus credit cards was the most effective in changing consumer behavior. When this disclosure was compared to a disclosure that displayed only the dollar cost of repeated payday loans, the take-up of future payday loans was reduced by 11 percent in the subsequent four months. Accompanying survey data suggest that the effect of the disclosures was stronger for customers who did not have a college degree and had higher self-control (on two self-reported measures).

Mutual Fund Disclosures

Mutual funds play an important role in U.S. household finances and retirement planning. According to the Investment Company Institute (ICI), 28 percent of U.S. retirement funds are invested in mutual funds. The U.S. mutual fund industry held \$15 trillion in assets at the end of 2013, accounting for 22 percent of household financial assets (ICI, 2015).

The Securities and Exchange Commission (SEC) regulates mutual funds and requires mutual fund companies to provide a fund's prospectus to all shareholders. The prospectus includes information on the fund's investment objectives or goals, investment strategies, risks, fees and expenses, past performance, and the fund's investment managers and advisers. It's a very challenging task for an average investor to navigate through the statutory prospectuses mutual finds distribute. The SEC commented that "the language of prospectuses is complex and legalistic, and the presentation formats make little use of graphic design techniques that would contribute to readability" (SEC, 2007).

A series of GAO reports and testimony in front of the House and Senate noted that disclosures pertaining to mutual funds often exceed those provided for other financial products, and acknowledged that additional disclosure would incur additional costs; nevertheless,

-

⁶ Payday loans are small-dollar, short-term, unsecured lending to borrowers typically experiencing cash flow difficulties (FDIC, 2003).

increased transparency is necessary if investors are to understand the potential conflicts of interest that their financial advisers face and be able to make informed decisions about how to invest their money, particularly when navigating complex and changing schedules of fees. GAO also notes that more transparency could promote greater price competition in the industry (GAO, 2003a; GAO, 2003b; GAO, 2004). In 2004, the GAO recommended further regulatory action by the SEC on mutual funds, though some in the industry and press expressed concern about potential overregulation (if some reforms created costs without clear benefits) and unequal treatment of mutual funds compared with other financial instruments ("A Rash of Rules," 2004).

On December 14, 2007, the SEC adopted the "Summary Prospectus Rules" aiming to simplify mutual fund prospectuses and "improve investors' ability to make informed investment decisions and, therefore, lead to increased efficiency and competitiveness of the U.S. capital markets" (SEC, 2007, Release No. 33-8861). With the new rules, the mutual funds have the option to send investors a three- or four-page Summary Prospectus.⁷

Beshears et al. (2011) examined differences in investor behavior between those who received the Summary Prospectus and those who received the longer statutory prospectus. They conducted a randomized experiment in which 186 subjects were asked to allocate two portfolios: one among four actively managed equity mutual funds, and one among four actively managed bond mutual funds. Participants were paid depending on how well their portfolios actually performed in the subsequent one-month or one-year period after the experimental session. Participants were randomly assigned to one of three conditions: one group received the funds' statutory prospectuses only, one group received the funds' Summary Prospectuses only, and the third group received the funds' Summary Prospectuses but with the option to request the statutory prospectuses. They found that participants who received the Summary Prospectus spent less time on investment decisions than those who received the statutory prospectus, but the authors found no significant differences in portfolio choices across treatments. Likewise, they did not find any significant differences across treatments in the loads, expense ratios, fees, and historical performance of the funds that participants chose.

In 2014, The SEC issued a Guidance Update on the Summary Prospectus, because it found that even though some funds' prospectuses are clear and concise, there are "a significant number of prospectuses, however, in which disclosure remains complex, technical and duplicative" (SEC, 2014). The new guidance highlights certain rules such as using plain English, focusing on the key information, and avoiding cross-references.

Haslem (2006) argues that another challenge for effective disclosure in the mutual fund industry is that readily disclosed fees do not necessarily reflect the true costs associated with the fund. Managerial fees are included in the benchmark "expense ratio," but other costs to the fund,

_

⁷ The statutory prospectus can run hundreds of pages long.

including brokerage commissions and some other implicit costs of trading,⁸ are not included. Haslem argues that reasonable estimates for these additional costs, using portfolio turnover to calculate expected values for the "other" costs, would give a potentially more useful "total expense ratio"; he shows how Vanguard's 500 Index fund (with relatively high rate of return and comparatively low turnover) performs particularly well when its rate of return is compared with its total expense ratio. He argues that higher total expense ratios of other funds should seem unreasonable to investors unless they are producing significantly higher returns, which is rare. Similar points were raised in a report by Zero Alpha Group (described in "Analysis Finds Fees Disclosed," 2004), showing that the highest-turnover funds may have hidden trading costs more than twice as high as published expense ratios, and that passively managed index funds (with low turnover and fees) will often outperform actively managed funds. For a sample of ten highturnover funds, the implicit trading costs averaged 1.91 percent; in one of the most egregious cases, the PBHG Large Cap Fund had an estimated implicit cost (due to turnover) of 4.27 percent and brokerage commissions of 3.16 percent, while publishing an expense ratio of only 1.16 percent. A recent study conducted by the North American Securities Administrators Association (NASAA) found a wide disparity in how fees were disclosed and concluded that disclosures may lose their effectiveness when hidden in small print, incorporated in lengthy account opening documents, or remaining opaque about the services provided (NASAA, 2014).⁹

While mandatory disclosure could provide (at some cost) a greater amount of information and transparency for investors, this may not be a full solution if investors do not understand or use this information correctly. Tkac (2004) believes that a lack of voluntary disclosure in the mutual fund industry may be a sign that investors do not value the information enough to be willing to pay for it. Palmiter and Taha (2008) survey academic literature and find that investors do not have a grasp of the data available to them, paying insufficient attention to data on expenses, risk profiles, and basic characteristics of funds, and instead focus most of their attention on historical rates of return (a strategy for which there is minimal empirical support). They conclude that disclosure may not be enough, and that regulators should pay more attention to research on the characteristics and capabilities of investors when formulating policy.

One special case of mutual fund disclosure is the purchase of mutual funds through banks. As banks are usually associated with fund security through the Federal Deposit Insurance Corporation (FDIC) insurance, the "Interagency Statement on Retail Sales of Nondeposit Investment Products" (1994) and the Gramm-Leach-Bliley Act (1999) require banks that sell mutual funds to disclose to investors that such funds are not guaranteed by the FDIC nor any government agency. There is limited and mixed evidence on the effectiveness of this disclosure.

-

⁸ In particular, Haslem draws attention to the effect of bid-ask spreads in trading—if the 'true' value of a particular asset is the midpoint between the bid and ask prices, then the distance between the trade price and that midpoint is an implicit cost of trading.

⁹ The study also found that in the outgoing transfer fees context, markups were routinely in the 100 percent to 280 percent range, perhaps in part due to consumer confusion about the fees being charged.

An AARP survey (1994) found that most bank consumers were unaware of the risks and fees of the uninsured investments sold by the banks. However, a Consumer Bankers Association survey conducted in 1994 (cited in Alexander, Jones, and Nigro, 2001) reported that few customers held the misconception that bank issued mutual funds were insured by the FDIC.

Discussion

In the financial decisionmaking areas that we focus on in this section—mortgages, credit cards and payday loans, and mutual funds—recent regulation of disclosure has all focused on simplifying disclosures. Research has shown that the longer, more detailed disclosure documents have not been effective at helping consumers make informed choices in selecting mortgages, credit cards, and mutual funds, due to either limited attention or limited understanding of the material itself.

Simpler disclosures, particularly for mortgages and credit cards, have shown promise in helping consumers make informed decisions, but research shows that even these simplified disclosures may still be too lengthy and complex for the average consumer. In preliminary research, the new Loan Estimate forms for mortgages have been shown to be easier for consumers to understand. However, because the new Loan Estimate forms have not yet been implemented, the impact of the forms on mortgage borrowers has not been fully evaluated. Simplified credit card disclosures as mandated by the CARD Act have been shown to influence credit card borrower behavior and understanding. In particular, the minimum payment warning has been shown to have a small impact on repayment behavior. One possible explanation for the small impacts on behavior is that the disclosures do not reach consumers—the disclosures are provided with the paper credit card statement, but more and more consumers pay their credit card bills online—and even when they do reach consumers, the financial terminology is difficult for consumers to understand. Summary Prospectuses for mutual funds have been shown to reduce the decision time, but not significantly affect consumer choice. This may be because although simplified, they are still relatively lengthy and do not present complex information, such as information on fees, in a uniform, straightforward manner.

5. Best Practices in Designing Effective Disclosures

In this section, we review best practices in designing effective financial disclosures. We draw from research on disclosures in other fields as well, especially when they address issues common to designing all disclosures, such as reading level or length.

Guidelines from Regulations

The Clear and Conspicuous Standards of the Federal Trade Commission

In 1970, concerned with the lack of adequate and accurate information in commercials, the Federal Trade Commission (FTC) issued the Clear and Conspicuous standard for advertising disclosure, including financial product advertising, such as credit card commercials.

The FTC proposed eight standards to improve disclosure quality, including presenting the disclosure in dual modes and using sufficient font size and readability level. The relevance of each standard depends on the particular product or service, and the context in which the disclosure is presented.

Results from tests of the impact of the FTC's standards on consumer comprehension have been mixed. In an online experiment in Thomas, Fowler, and Kolbe (2011), 505 college students read one of 12 versions of a credit card advertisement containing disclosure information. The experiment is designed so that three FTC standards are tested. Subjects were presented with disclosures in one of two modes: In the single modality condition, the disclosure is presented visually only, whereas in the dual modality condition, the disclosure is presented visually while a male voice reads it aloud. Students also received disclosures with one of two readability levels, either with jargon, or jargon free, and with one of three distraction levels, either a blank white background, a still image as background, or a background with dynamic music and scene changes. Their results suggest that dual modality and easy readability improve subjects' comprehension only when there is minimum distraction. In other words, when there is background distraction, as there often is in advertising, jargon-free disclosures presented in dual modes do not increase comprehension when compared to single-mode disclosures with jargon.

SEC's Plain English Initiative

In 1998, the SEC published "A Plain English Handbook: How to Create Clear SEC Disclosure Documents" (SEC, 1998). The 83-page guideline was designed to urge disclosers to generate simpler and clearer documents for investors. The handbook suggests that in order to design effective disclosure, the issuers need to understand both audience and content, assemble experienced drafters to weigh the length of the planned document, and adopt certain effective

writing styles (e.g., use active voice, simple words, short sentences, and common terms). The handbook also provides tips and suggestions on procedures to follow when designing disclosures, evaluation methods to evaluate the disclosures, and useful tools such as a readability formula and style checker.

FTC's .Com Disclosure Guidelines

Many policies and guidelines were designed based on paper disclosures, while people increasingly make decisions online. In 2013, the FTC published an online advertising disclosure guideline in which the agency proposed 16 suggestions on how to make a disclosure clear and conspicuous ¹⁰ (FTC 2013, pp. ii–iii). Several suggestions are specific to internet disclosures: Take account of the various devices and platforms on which the disclosures are viewed; make use of obvious and well-labeled links in a consistent hyperlink style for disclosures that cannot be incorporated in a space-constrained ad; monitor click-through rates of disclosure links; avoid scrolling or use visual cues to prompt scrolling; repeat disclosure at different decision points and on different web pages; and prominently display disclosures and limit distraction from other elements on the web page.

Regulatory Guideline on the Timing of Disclosure

There are regulatory guidelines on the timing of disclosure to ensure that customers receive the information in a timely fashion and have enough time to process it. For example, in mortgage disclosures, the GFE Statement is required within three business days of the borrower's application, and a preliminary copy of a HUD-1 Settlement Statement is required if the borrower requests it 24 hours before closing. The 2009 CARD Act requires that credit card bills be due on the same date each month, and the statement be provided at least 21 days before the due date. However, it remains largely unknown whether disclosure provided in advance is more effective in influencing decisions than disclosure provided at the moment a financial decision is being made.

Lessons Learned from Effective and Ineffective Disclosures

Disclosure is effective only when the receivers pay attention to the information, have the capacity to interpret it, and are willing to incorporate it in their decisionmaking process. However, many customers do not read or pay little attention to the disclosures (Nash, 2009), lack the basic financial literacy to understand the terms (J. D. Powers, 2013), and fail to include the disclosed information at the decision point (Sovern, 2010). It is challenging to convey complex and unfamiliar financial and legal concepts to an average American consumer. In this section, we

_

¹⁰ The FTC.com Disclosure also provides guidelines on other perspectives besides how to make clear and conspicuous disclosures, such as "the same consumer protection laws that apply to commercial activities in other media apply online, including activities in the mobile marketplace," etc.

draw upon lessons learned from both successful and unsuccessful disclosure practice in the finance industry as well as other decisionmaking domains.

The Gramm-Leach-Bliley Act of 1999 (GLBA) required financial institutions to provide a privacy notice to all their customers. The disclosures sent by the financial institutions subsequently were "lengthy, confusing, written in a highly legalistic style and generally incomprehensible" (Garrison et al., 2012). In response to the frustration and critique of consumers, media, and members of Congress, six federal agencies launched a consumer research project that had two phases: using qualitative methods to iteratively develop a new prototype notice, and a quantitative study to evaluate the prototype notice in comparison with other privacy notices. Garrison et al. (2012) identified a list of lessons learned from the research project to develop disclosures that consumers pay attention to, understand, and use in their decisionmaking:

- Use plain English
- Provide a context for concepts unfamiliar to consumers
- Presenting information in tables can be helpful, but not all disclosed information is better presented in tables
- Pay attention to design features: contextual framing, visual hierarchy, information sequencing, font choice, white space, etc.
- Layer information carefully: Identify key pieces of information and locate them together in a highly visible place and format; people pay more attention to information on the first page
- Be consumer driven: Study and understand consumers, involve consumers in development of disclosures, be clear about the goal (whether to inform or to encourage actions), provide additional education and supplementary material if necessary
- Involve responsible policy staff
- Use both qualitative and quantitative methods when developing the disclosures
- Exploit synergies: Prioritize information in an accessible format; seek resolution within the confines of the laws and regulations.

Fung, Graham, and Weil (2007) provide in-depth assessment of eight mandatory disclosure policies. Based on empirical evidence of numerous studies to evaluate the eight kinds of disclosures, the authors conclude that some policies (corporate financial, mortgage lending, and restaurant hygiene disclosures) are effective, some policies (patient safety and plant closing) have failed, while other policies (nutritional labeling, toxic releases, and work hazard disclosures) may be moderately effective. The authors identify ten principles for policymakers in crafting effective policies:

- Provide easy-to-use information: Information needs to be able to be conveniently embedded in the normal decision process
- Strengthen user groups, such as labor unions or consumer groups, as they can help to monitor, maintain, and improve disclosures
- Help disclosers understand how customer behavior can be impacted by disclosure
- Design disclosures so that disclosers perceive benefits from transparency
- Standardize metrics for information accuracy and product comparability

- Match information content and formats to users' levels of attention and comprehension
- Provide periodic analysis, feedback, and revision to avoid outdated systems
- Impose sanctions on disclosers for noncompliance
- Strengthen enforcement of compliance with disclosure rules
- Leverage other regulatory systems besides disclosure policies.

Lastly, evidence from the medical field suggests that advances in technology may provide new opportunities to effectively provide disclosures. Madathil et al. (2013) find that presenting informed consent forms on iPads instead of paper can improve patient understanding of the content of the forms. Tait et al. (2012) ran a lab study in which participants were randomized to view risk/benefit information of a medication on an iPad in one of four formats: text, pie chart, bar graph, and pictograph. The authors found that when participants viewed the information in the format they prefer, their understanding and satisfaction were significantly better than if there was not a match between the preferred format and the view format. An implication is that tailoring may be an important factor in improving disclosure effectiveness and can be implemented by human-computer interaction at the disclosure point. That is, instead of the traditional "one solution fits all" messages, computers can help issuers customize disclosures.

Discussion

One common factor in discussion of effective disclosures is simplicity—namely, using plain English and keeping technical jargon to a minimum. Generally, however, what is effective may be specific to the decision at hand. Using tables and graphs may aid in understanding, but the suitability of using them depends on the topic as well as the audience. New technology, such as tablet computers and other mobile devices, may become an important tool in helping design more effective disclosures that can be customized to individual customers.

Fung, Graham, and Weil (2007) suggest that disclosure can be an effective policy tool when there is an information gap that disclosure can fill. For example, warning people about a particular risk that they would otherwise not be aware of can help them make more informed decisions and take appropriate actions when facing the risk. On the other hand, if there is no defined information gap meriting government intervention, disclosure is less effective. One such example is the case of international labeling of genetically modified goods, because nations do not agree about whether genetic engineering creates a public safety problem. Fung, Graham, and Weil also suggest that disclosure is more effective when the decision arena is not too complex to communicate. In the example of disclosing of workplace hazards, the complexity of risk exposure information was a major barrier for the policy's effectiveness. Another condition for the disclosure to be effective is that consumers have the desire and capability to improve their decisions. Information will be ignored unless the consumers have more desire to incorporate it when making decisions. For example, public information is available on the likelihood of natural disasters by city, but residents rarely incorporate such information when considering a move.

Lastly, the extent to which disclosure can positively affect the discloser's behavior is contingent on the discloser's capacity to reduce risks or improve performance. For example, environmental or nutritional labeling may not affect a producer's behavior, as the ability to reduce toxic pollution or remove harmful fats from foods depends on the availability of substitutes and related technology.

6. Summary

In theory, consumer disclosures can reduce the information asymmetries present in the financial services market. Disclosures can provide investors with valuable information so that they can make informed decisions. In this report, we reviewed the literature on consumer disclosures in the financial industry, focusing on research on the effectiveness of disclosure in providing decision support to investors. Our review of existing studies indicates that disclosure, particularly disclosure used in isolation, may not provide sufficient support in helping investors make more informed decisions.

Disclosure is a key component of the regulatory regime in the U.S. financial advice market as well as in many financial advice markets around the globe. In markets in which advisers are permitted to receive commission, advisers are frequently required to disclose the nature and amount of compensation they receive as a result of the advice provided. In markets in which commissions are banned, advisers are still frequently required to explicitly disclose all costs related to the advice provided and whether it considers all available products.

In Chapter 3, we reviewed literature on the impact of disclosures on conflicts of interest. The available research suggests that when conflicts of interest in financial advice are disclosed, many consumers fail to adjust their behavior sufficiently, if at all. Furthermore, disclosure may in fact have the opposite effect: Consumers may feel a "burden of disclosure" to follow the advice, and advisers may respond to the disclosure by providing even more biased advice, resulting in decreased welfare for consumers. In other areas of financial decisionmaking such as mortgages, credit cards and payday loans, and mutual funds, disclosure has long been used as a policy tool. The trends in regulation on disclosure in these areas have all focused on simplifying disclosures. Research has shown that the longer, more detailed disclosure documents have not been effective at helping consumers make informed choices in selecting mortgages, credit cards, and mutual funds, due to either limited attention or limited understanding of the material itself. Simpler disclosures, particularly for mortgages and credit cards, have shown promise in helping consumers make informed decisions, but research shows that even these simplified disclosures may still be too lengthy and complex for average consumers.

In Chapter 5, we examined the instances in which disclosure can be most effective. Simplicity in disclosures is a necessary but not sufficient condition in designing effective disclosure. Disclosure tends to be more effective when the decision arena is not too complex to communicate, and when consumers have the desire and capability to improve their decisions. However, research indicates that disclosure alone, even simplified disclosure, may not be effective at improving financial decisionmaking. Given that many consumers have low levels of financial capability, disclosure is likely to be most effective when used in conjunction with other policy tools.

References

- Agarwal, S., S. Chomsisengphet, N. Mahoney, and J. Stroebel, "Regulating Consumer Financial Products: Evidence from Credit Cards," *The Quarterly Journal of Economics*, 2014, pp. 111–164.
- Alexander, G. J., J. D. Jones, and P. J. Nigro, "Does Mutual Fund Disclosure at Banks Matter? Evidence from a Survey of Investors," *The Quarterly Review of Economics and Finance*, Vol. 41, No. 3, 2001, pp. 387–403.
- "Analysis Finds Fees Disclosed by Funds Understate 'True Costs'," *Managing 401(k) Plans*, Vol. 4, April 1, 2004, pp. 1–14.
- "A Rash of Rules," *Economist*, June 24, 2004, pp. 75–76.
- Australian Securities and Investments Commission, *Regulatory Guide 245: Fee Disclosure Statements*, March 1, 2013.
- BaFin (Federal Financial Supervisory Authority), home page, undated. As of August 10, 2015: http://www.bafin.de/EN/Homepage/homepage_node.html;jsessionid=EDD1C165C64748BE F09AE058F604A6F4.1 cid290
- Bertrand, M., and A. Morse, "Information Disclosure, Cognitive Biases, and Payday Borrowing," *The Journal of Finance*, Vol. 66, No. 6, December 2011, pp. 1865–1893.
- Beshears, J., J. J. Choi, D. Laibson, and B. C. Madrian, "How Does Simplified Disclosure Affect Individuals' Mutual Fund Choices?" In *Explorations in the Economics of Aging*, ed. David A. Wise, Chicago: University of Chicago Press, 2011, pp. 75–96.
- Burke, J., and A. Hung, "Financial Advice Markets: A Cross Country Comparison," Santa Monica, Calif.: RAND Corporation, working paper, 2015.
- Cain, D. M., G. Loewenstein, and D. A. Moore, "The Dirt on Coming Clean: Perverse Effects of Disclosing Conflicts of Interest" *Journal of Legal Studies*, Vol. 34, No. 1, January 2005, pp. 1–25.
- Cain, D. M., G. Loewenstein, and D. A. Moore, "When Sunlight Fails to Disinfect: Understanding the Perverse Effects of Disclosing Conflicts of Interest, *Journal of Consumer Research*, Vol. 37, No. 5, February 2011, pp. 836–857.
- CFPB—See Consumer Financial Protection Bureau.
- Chater, N., S. Huck, and R. Inderst, *Consumer Decision-Making in Retail Investment Services: A Behavioural Economics Perspective, Final Report*, November 2010.

- Church, B. K., and X. J. Kuang, "Conflicts of Interest, Disclosure, and (Costly) Sanctions: Experimental Evidence," *Journal of Legal Studies*, Vol. 38, No. 2, 2009, pp. 505–532.
- Collins, J. M., "How Good Is the GFE? How Truthful Is the TILA? Comparing Mortgage Loan Disclosures to Settlement Documents," *Housing Policy Debate*, Vol. 21, No. 4, September 2011, pp. 565–583.
- Consumer Financial Protection Bureau, "Testing 'Know Before You Owe' Mortgage Forms," November 20, 2013a. As of August 12, 2015:
 - http://files.consumerfinance.gov/f/201311_cfpb_factsheet_kbyo_testing.pdf
- ———, "CFPB Consumer Laws and Regulations: TILA," June 2013b. As of August 12, 2015: http://files.consumerfinance.gov/f/201306_cfpb_laws-and-regulations_tila-combined-june-2013.pdf
- ——, "CARD Act Report: A Review of the Impact of the CARD Act on the Consumer Credit Card Market," October 1, 2013c. As of August 12, 2015: http://files.consumerfinance.gov/f/201309 cfpb card-act-report.pdf
- Council of Economic Advisers, *The Effects of Conflicted Investment Advice on Retirement Savings*, 2015. As of August 12, 2015: https://www.fi360.com/offers/webinars/the-effects-of-conflicted-investment-advice-on-retirement-savings
- Durkin, T. A., "Credit Card Disclosures, Solicitations, and Privacy Notices: Survey Results of Consumer Knowledge and Behavior," *Federal Reserve Bulletin*, 2006, pp. A109–A121.
- FCA—See Financial Conduct Authority.
- FDIC—See Federal Deposit Insurance Corporation.
- Financial Conduct Authority, "Supervising Retail Investment Advice: Delivering Independent Advice," TR14/5, March 2014a. As of August 12, 2015:
 - https://www.fca.org.uk/static/documents/thematic-reviews/tr14-05.pdf
- , "Supervising Retail Investment Firms: Being Clear About Adviser Charges and Services," TR14/6, April 7, 2014b. As of August 12, 2015:
 - http://www.fca.org.uk/news/tr14-6-supervising-retail-investment-firms
- ——, "Retail Investment Advice: Adviser Charging and Services," TR14/21, December 16, 2014c. As of August 12, 2015:
 - https://www.fca.org.uk/news/tr14-21-retail-investment-advice
- ——, "Retail Distribution Review Thematic Work: Publications and Supporting Material," 2015. As of August 10, 2015:
 - http://www.fca.org.uk/firms/firm-types/sole-advisers/rdr

- Federal Deposit Insurance Corporation, "An Update on Emerging Issues in Banking: Payday Lending," January 29, 2003. As of August 12, 2015: https://www.fdic.gov/bank/analytical/fyi/2003/012903fyi.html
- Federal Financial Supervisory Authority, *BaFin Quarterly*, First Quarter Edition, 2010. As of August 14, 2015: http://www.bafin.de/SharedDocs/Downloads/EN/Mitteilungsblatt/Quarterly/bq1101.pdf?__bl ob=publicationFile
- Federal Trade Commission, ".com Disclosures: How to Make Effective Disclosures in Digital Advertising," press release, March 2013. As of August 12, 2015: https://www.ftc.gov/sites/default/files/attachments/press-releases/ftc-staff-revises-online-advertising-disclosure-guidelines/130312dotcomdisclosures.pdf
- Foster, K., E. Meijer, S. D. Schuh, and M. A. Zabek, "The 2009 Survey of Consumer Payment Choice," Federal Reserve Bank of Boston Public Policy Discussion Paper, Issue 11-1, April 1, 2011.
- FTC—See Federal Trade Commission.
- Fung, A., M. Graham, and D. Weil, *Full Disclosure: The Perils and Promise of Transparency*, Cambridge, UK: Cambridge University Press, 2007.
- GAO—See U.S. Government Accountability Office.
- Garrison, L., M. Hastak, J. M. Hogarth, S. Kleimann, and A. S. Levy, "Designing Evidence-Based Disclosures: A Case Study of Financial Privacy Notices," *Journal of Consumer Affairs*, Vol. 46, No. 2, Summer 2012, pp. 204–234.
- George Washington University, School of Public Health and Health Services, "Pharmaceutical Marketing Expenditures in the District of Columbia, 2010," March 2, 2012. As of August 12, 2015:
 - $http://doh.dc.gov/sites/default/files/dc/sites/doh/publication/attachments/pharmaceutical_marketing_expenditures_in_the_district_of_columbia_2010.pdf$
- Gibbons, R. V., F. J. Landry, D. L. Blouch, D. L. Jones, F. K. Williams, C. R. Lucey, and K. Kroenke, "A Comparison of Physicians' and Patients' Attitudes Toward Pharmaceutical Industry Gifts" *Journal of General Internal Medicine*, Vol. 13, No. 3, March 1998, pp. 151–154.
- Green, K. C., and J. S. Armstrong, "Evidence on the Effects of Mandatory Disclaimers in Advertising," *Journal of Public Policy and Marketing*, Vol. 31, No. 2, Fall 2012, pp. 293–304.
- Haslem, J. A., "Assessing Mutual Fund Expenses and Transaction Costs," *Journal of Investing*, Vol. 15, No. 4, Winter 2006, pp. 52–56.

- ICI—See Investment Company Institute.
- Inderst, R., and M. Ottaviani, "Competition Through Commissions and Kickbacks," *American Economic Review*, Vol. 102, No. 2, 2012a, pp. 780–809.
- , "Financial Advice," *Journal of Economic Literature*, Vol. 50, No. 2, 2012b, pp. 494–512.
- ——, "How (Not) to Pay for Advice: A Framework for Consumer Financial Protection," *Journal of Financial Economics*, Vol. 105, No. 2, 2012c, pp. 393–411.
- "Integrated Mortgage Disclosures Under the Real Estate Settlement Procedures Act (Regulation X) and the Truth in Lending Act (Regulation Z)," Federal Register Unified Agenda 3170-AA19, 2013. As of August 12, 2015:

 https://www.federalregister.gov/regulations/3170-AA19/integrated-mortgage-disclosures-under-the-real-estate-settlement-procedures-act-regulation-x-and-the
- "Interagency Statement on Retail Sales of Nondeposit Investment Products," Office of the Comptroller of the Currency, Bulletin 1994–13, February 15, 1994.
- Investment Company Institute, "2014 Investment Company Fact Book," 2015. As of August 13, 2015:
 - http://www.icifactbook.org/2014/
- J. D. Power, "American Express Ranks Highest in Customer Satisfaction for a Seventh Consecutive Year," August 22, 2013. As of August 13, 2015: http://www.jdpower.com/zh-hans/node/2918
- Jones, L. E., C. Loibl, and S. L. Tennyson, "The Effects of CARD Act Disclosures on Consumers' Use of Credit Cards," working paper, June 2012.
- Klass, J. L., "Investment Adviser Conflicts of Interest Disclosures," Investment Adviser Association Annual Compliance Workshop, New York, New York, October 27, 2008.
- Lacko, J. M., and J. K. Pappalardo, "Improving Consumer Mortgage Disclosures—An Empirical Assessment of Current and Prototype Disclosure Forms," Washington, D.C.: Federal Trade Commission, Bureau of Economics, 2007.
- ——, "The Failure and Promise of Mandated Consumer Mortgage Disclosures: Evidence from Qualitative Interviews and a Controlled Experiment with Mortgage Borrowers," *American Economic Review*, Vol. 100, No. 2, May 2010, pp. 516–521.
- Loewenstein, G., D. M. Cain, and S. Sah, "The Limits of Transparency: Pitfalls and Potential of Disclosing Conflicts of Interest," *The American Economic Review*, Vol. 101, No. 3, May 2011, pp. 423–428.

- Madathil, K. C., R. Koikkara, J. Obeid, J. S. Greenstein, I. C. Sanderson, K. Fryar, J. Moskowitz, and A. K. Gramopadhye, "An Investigation of the Efficacy of Electronic Consenting Interfaces of Research Permissions Management System in a Hospital Setting," *International Journal of Medical Informatics*, Vol. 82, No. 9, 2013, pp. 854–863.
- MAS—See Monetary Authority of Singapore.
- Monetary Authority of Singapore, "Financial Advisers Act (Cap. 110): Notice on Information to Clients and Product Information Disclosure," Notice No. FAA-N03, October 1, 2002.
- ——, "Mystery Shopping Survey Findings," 2012.

 http://www.mas.gov.sg/news-and-publications/pressreleases/2012/~/media/MAS/Regulations%20and%20Financial%20Stability/Regulations%20
 Guidance%20and%20Licensing/Financial%20Advisers/Others/MSFindings2012.ashx
- NASAA—See North American Securities Administrators Association.
- Nash, Jeff, "2 of 3 Participants Don't Read Info," *Pensions and Investments*, Vol. 37, No. 21, 2009, pp. 2–42.
- Navarro-Martinez, D., L. C. Salisbury, K. N. Lemon, N. Stewart, W. J. Matthews, and A. J. L. Harris, "Minimum Required Payment and Supplemental Information Disclosure Effects on Consumer Debt Repayment Decisions," *Journal of Marketing Research*, Vol. 48, 2011, pp. S60–S77.
- North American Securities Administrators Association, "Broker-Dealer Fee Survey," NASAA Broker-Dealer Investment Products and Services Project Group, 2014.
- Palmiter, A. R., and A. E. Taha, "Mutual Fund Investors: Sharp Enough?" *Journal of Financial Transformation*, Vol. 24, January 2008, pp. 113–121. As of August 13, 2015: http://www.researchgate.net/publication/46557869_Mutual_Fund_Investors_Sharp_Enough
- Pearson, S. D., K. Kleinman, D. Rusinak, and W. Levinson, "A Trial of Disclosing Physicians' Financial Incentives to Patients," *Archives of Internal Medicine*, Vol. 166, No. 6, March 2006, pp. 623–628.
- Sah, S., G. Loewenstein, and D. M. Cain, "The Burden of Disclosure: Increased Compliance with Distrusted Advice," *Journal of Personality and Social Psychology*, Vol. 104, No. 2, 2013, pp. 289–304.
- Sah, S., and G. Loewenstein, "Nothing to Declare: Mandatory and Voluntary Disclosure Leads Advisers to Avoid Conflicts of Interest," *Psychological Science*, Vol. 25, No. 2, February 2014, pp. 575–584.

- Salisbury, L. C., "Minimum Payment Warnings and Information Disclosure Effects on Consumer Debt Repayment Decisions," *Journal of Public Policy and Marketing*, Vol. 33, No. 1, Spring 2014, pp. 49–64.
- SEC—See U.S. Securities and Exchange Commission.
- Soll, J. B., R. L. Keeney, and R. P. Larrick, "Consumer Misunderstanding of Credit Card Use, Payments, and Debt: Causes and Solutions," *Journal of Public Policy and Marketing*, Vol. 32, No. 1, May 2013, pp. 66–81.
- Sovern, J., "Preventing Future Economic Crises Through Consumer Protection Law or How the Truth in Lending Act Failed the Subprime Borrowers," *Ohio State Law Journal*, Vol. 71, No. 4, 2010, pp. 761–844.
- ——, "Fixing Consumer Protection Laws So Borrowers Understand Their Payment Obligations," *Journal of Consumer Affairs*, Vol. 48, No. 1, Spring 2014, pp. 17–33.
- Stewart, N., "The Cost of Anchoring on Credit-Card Minimum Repayments," *Psychological Science*, Vol. 20, No. 1, January 2009, pp. 39–41.
- Tait, A. R., T. Voepel-Lewis, C. Brennan-Martinez, M. McGonegal, and R. Levine, "Using Animated Computer-Generated Text and Graphics to Depict the Risks and Benefits of Medical Treatment," *American Journal of Medicine*, Vol. 125, No. 11, November 2012, pp. 1103–1110.
- Thomas, V., K. Fowler, and R. H. Kolbe, "The Implications of the FTC's Clear and Conspicuous Standards for the Communication of Credit Card Information to Young Consumers," *Journal of Financial Services Marketing*, Vol. 16, No. 3–4, December 2011, pp. 195–209.
- Tkac, Paula A., "Mutual Funds: Temporary Problem or Permanent Morass?" *Federal Reserve Bank of Atlanta Economic Review*, Vol. 89, No. 4, 2004, pp. 1–21.
- U.S. Census Bureau, "Statistical Abstract of the United States: 2012," Table 1188, Washington, D.C.: U.S. Department of Commerce, 2012. As of August 13, 2015: http://www.census.gov/compendia/statab/
- U.S. Department of the Treasury, "Financial Regulatory Reform: A New Foundation," Washington, D.C., 2009.
- U.S. Government Accountability Office, "Mutual Funds: Information on Trends in Fees and Their Related Disclosure," GAO-03-551T, March 12, 2003a. As of August 13, 2015: http://www.gao.gov/products/GAO-03-551T

